Author's response to reviews

Title: Geriatric rehabilitation of stroke patients in nursing homes: a study protocol

Authors:

Monica Spruit- van Eijk (m.spruit-vaneijk@elg.umcn.nl)
Bianca I Buijck (b.buijck@elg.umcn.nl)
Sytse U Zuidema (s.zuidema@elg.umcn.nl)
Frans LM Voncken (frans.voncken@zorgboog.nl)
Alexander CH Geurts (a.geurts@reval.umcn.nl)
Raymond TCM Koopmans (r.koopmans@elg.umcn.nl)

Version: 3 Date: 17 March 2010

Author's response to reviews: see over
Dear Madam, Sir,

We enclosed our revised study protocol ‘Geriatric Rehabilitation of Stroke Patients in Nursing Homes’ by Monica Spruit- van Eijk, Bianca I. Buijck, Sytse U. Zuidema, Frans L.M. Voncken, Alexander C.H. Geurts, and Raymond T.C.M. Koopmans.

We have revised our manuscript and made the following changes:
1. The author list has been revised to ‘First name Initial Last name’.
2. The affiliations have been written in full.
3. The shade has been removed from the tables.
4. Word count and table count has been removed from the title page.
5. The authors’ qualifications have been removed from the title page.
6. ‘No commercial or financial involvements’ has been removed from the title page.
7. Table placement in text has been removed.
8. We checked the manuscript for typographical errors.

Thank you for your comments.

We are looking forward to your reply.

Kind regards,

Monica Spruit- van Eijk, elderly care physician
Department of primary and community care
117 VPHG Radboud University Nijmegen, Medical Centre
PO Box 9101
6500 HB Nijmegen The Netherlands
e-mail: M.Spruit-vanEijk@elg.umcn.nl
telephone: +31 24 3618834
fax: +31 24 3619558